Author’s response to reviews

Title: Protocol for Northern Ireland Caries Prevention in Practice Trial (NIC-PIP)
trial: a randomised controlled trial to measure the effects and costs of a dental
cares prevention regime for young children attending primary care dental
services

Authors:

martin Tickle (martin.tickle@manchester.ac.uk)
Helen V Worthington (Helen.Worthington@manchester.ac.uk)
Michael Donaldson (Michael.Donaldson@hscni.net)
Seamus Killough (sskilldent@hotmail.com)
Ciaran O'Neill (ciaran.oneill@nuigalway.ie)
Grainne Crealey (Grainne.crealey@crsc.n-i.nhs.uk)
Matt Sutton (matt.sutton@manchester.ac.uk)
Solveig Noble (Solveig.Noble@northerntrust.hscni.net)
Margaret Greer (MargaretGreer@crsc.n-i.nhs.uk)
Milsom Keith (keith.milsom@hsthpcnhs.uk)

Version: 2 Date: 18 September 2011

Author’s response to reviews: see over
Dear Sir,

Subject: Formatting changes

Author list: Martin Tickle, Helen V Worthington, Michael Donaldson, Seamus Killough, Ciaran O'Neill, Grainne Crealey, Matt Sutton, Solveig Noble, Margaret Greer and Milsom Keith

Title: Protocol for Northern Ireland Caries Prevention in Practice Trial (NIC-PIP) trial: a randomised controlled trial to measure the effects and costs of a dental caries prevention regime for young children attending primary care dental services
Journal: BMC Oral Health
MS ID: 1164399276577464

I have made the following changes to the format of my manuscript, as requested so that the published article conforms to journal style.

Major revisions

Title: please ensure that the title is identical in the manuscript and the submission system - Confirmed

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' Amended

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution Amended

Abstract: please include a 'Discussion' section after your 'Methods/Design'. 'Discussion' section now included

Authors' contributions: Please use initials instead of full name. Amended

Figures: please upload your figures in the submission system and figure files. They are currently missing. Uploaded

Minor revisions

Graphics: please remove the envelope images from the title page. Removed

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. Removed

Structure: please change 'Methods' to 'Methods/Design' Amended

Borders: please remove all of these. Removed

Figures: please remove the figure titles from the text and place them in a 'Figure Legends' section at the end of the manuscript after the references. Amended

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. Checked

I hope these revisions meet with your approval. Thank you for your help with the manuscript

Best wishes,

Martin Tickle
Professor of Dental Public Health & Primary Care

University of Manchester